Author's response to reviews

Title: Contralateral extradural hematoma following decompressive craniectomy for acute subdural hematoma: case presentation

Authors:

Lucas C Meguins (lucascrociati@hotmail.com)
Gustavo B Sampaio (gustavo@hotmail.com)
Eduardo C Abib (eduardo@hotmail.com)
Rodrigo R Adry (rodrigo@hotmail.com)
Richam F Ellakkis (richam@hotmail.com)
Filipe W Ribeiro (filipe@hotmail.com)
Angelo L Masset (angelo@hotmail.com)
Dionei F Morais (dionei.fm@terra.com.br)

Version: 2
Date: 14 November 2013

Author's response to reviews:

We changed the 'Case Report' section header to 'Case Presentation'.
We changed the description of the patient's gender from 'man' to 'male'.
We included the Ethnicity of the patient in the Abstract and Case Presentation section.
We confirm that the family of the patient has given their consent for the case report to be published and included a Consent section in the manuscript stating that consent has been obtained.
We included an Authors' Contributions section at the end of the manuscript.
We included a Competing Interests section before the Reference list.